Author's response to reviews

Title: The cross-national epidemiology of social anxiety disorder: Data from the World Mental Health Survey Initiative

Authors:

Dan Stein (dan.stein@uct.ac.za)
Carmen Lim (c.lim@uq.edu.au)
Annelieke Roest (a.m.roest@umcg.nl)
Peter de Jonge (peter.de.jonge@rug.nl)
Sergio Aguilar-Gaxiola (aguilargaxiola@ucdavis.edu)
Ali Al-Hamzawi (ali_alhamzawi2000@yahoo.com)
Jordi Alonso (jalonso@imim.es)
Corina Benjet (cbenjet@gmail.com)
Evelyn Bromet (Evelyn.Bromet@stonybrookmedicine.edu)
Ronny Bruffaerts (ronny.bruffaerts@uzleuven.be)
Giovanni de Girolamo (gdegirolamo@fatebenefratelli.it)
Silvia Florescu (florescu.silvia@gmail.com)
Oye Gureje (oye_gureje@yahoo.com)
Josep Haro (jmharo@pssjd.org)
Meredith Harris (Meredith_Harris@qcmhr.uq.edu.au)
Yanling He (heyl2001@163.com)
Hristo Hinkov (h.hinkov@ncpha.government.bg)
Itsuko Horiguchi (itsukoh@nagasaki-u.ac.jp)
Chiyi Hu (chiyihu@126.com)
Aimee Karam (aimeekaram@hotmail.com)
Author’s response to reviews:

Dear Dr Recchioni,

Thank you for your email of 02 May 2017. We are delighted that the reviewers are satisfied.

As requested, we have added a list of competing interests, study approval statement, authors' contributions, and a conclusion header in the Discussion section.

We hope that the submission is now acceptable for publication.

Sincerely,

Dan Stein